

Society for the Psychological Study of Culture, Ethnicity, and Race

2020, Vol. 26, No. 4, 544-556 http://dx.doi.org/10.1037/cdp0000325

Postpartum Depressive Symptoms in Low-Income Latinas: Cultural and Contextual Contributors

Carolyn Ponting, Denise A. Chavira, Isabel Ramos, and Wendy Christensen University of California, Los Angeles Christine Guardino Dickinson College

Christine Dunkel Schetter University of California, Los Angeles

Objective: Using a conceptual model of postpartum depression risk in Latinas including both contextual and cultural stressors, we tested contributions to depressive symptom levels and trajectories over the course of 1 year following birth in a community sample of Latinas. *Method:* A multisite sample of low-income U.S.-born and foreign-born Latinas (n = 537; M age = 25.70) was interviewed on many topics including measures of stress and maternal health at 1, 6, and 12 months postpartum. Nested multilevel growth curve models were implemented to test associations of contextual stressors (poverty, domestic violence) with trajectories of depressive symptoms, adjusting for confounds. This model was compared to 1 that added cultural stress variables (everyday discrimination, foreign-born status, language preference, age at immigration) measured 1-month postpartum. *Results:* The best fitting model provided evidence for the independent effects of cultural and contextual stressors. Discrimination ($\beta = .13 SE = .02$, p = < .001) and domestic violence ($\beta = .39 SE = .09$, p = < .001) predicted trajectories with higher levels of depressive symptoms 1 month postpartum, but not linear change in symptoms over the year. *Conclusions:* The present study provides evidence that discrimination, a cultural factor, and domestic violence, a contextual factor, each predict higher levels of early postpartum depressive symptoms. Interventions addressing discrimination and maternal safety are recommended.

Public Significance Statement

This study highlights the importance of contextual and cultural stress in understanding postpartum depressive symptoms in low-income Latinas. Discrimination and domestic violence warrant further consideration in ethnic minority women when planning clinical and community interventions, especially in pregnancy when addressing stressors may prevent postpartum depressive symptoms.

Keywords: Latina, postpartum depression, discrimination, domestic violence, longitudinal

Postpartum depression is associated with adverse maternal and child outcomes (O'Hara & McCabe, 2013). Mothers with symptoms of depression following birth are more likely to have subsequent depressive episodes and to engage in hostile and unresponsive parenting behaviors (Brummelte & Galea, 2016). Their offspring have an increased risk for cognitive deficits, psychopathology, and poorer physical health from early childhood to adolescence (Field, 2010; Pearlstein, Howard, Salisbury, & Zlotnick, 2009). Estimates of the prevalence rates of postpartum depression are between 7 and 19% (Gavin et al., 2005), although differential rates by ethnicity are not well-established.

Nationally representative data describing rates of depression for women by ethnicity are mixed. Some show that Latinas have higher prevalence rates of depression than do White women (Bromberger, Harlow, Avis, Kravitz, & Cordal, 2004) and others that White women have higher rates (Guo, Robakis, Miller, & Butwick, 2018). Regarding postpartum depression, some studies report elevated symptoms of postpartum depression in Latinas compared to non-Latina Whites (Howell, Mora, Horowitz, & Leventhal, 2005) especially foreign-born Latinas (Fung & Dennis, 2010; Lucero, Beckstrand, Callister, & Sanchez Birkhead, 2012), while other studies report no ethnic differences (Liu & Tronick, 2012; Yonkers et al., 2001). Population-based surveys

This article was published Online First February 27, 2020.

Carolyn Ponting, Denise A. Chavira,
Isabel Ramos, and
Wendy Christensen, Department of Psychology, University of California, Los Angeles;
Christine Guardino, Department of Psychology, Dickinson College; Christine Dunkel Schetter, Department of Psychology, University of California, Los Angeles.

This work was made possible by the National Institute of Mental Health (NIH 5T32MH015750-38).

Correspondence concerning this article should be addressed to Carolyn Ponting, Department of Psychology, University of California, Los Angeles, 1183C Franz Hall, 502 Portola Plaza, Los Angeles, CA 90095. E-mail: cponting@ucla.edu

administered in the United States have shown that Latinas are two times more likely to be diagnosed with postpartum depression than White women, with rates of 7.8% versus 3.3%, respectively (Liu & Tronick, 2013). Further, Latinas with postpartum depression are significantly less likely to access mental health care when they experience symptoms (Chang, Tabet, Elder, Kiel, & Flick, 2016; Kozhimannil, Trinacty, Busch, Huskamp, & Adams, 2011), and have been identified as a group at heightened risk for perinatal depression (Muñoz et al., 2007).

The significance of this health disparity is magnified because Latinx are the largest ethnic minority group in the United States and account for 18% of the population (Pew Research Center, 2017). In 2016, Latina mothers gave birth to over 900,000 babies in the U.S, a birthrate 31% greater than the national average (Martin, Hamilton, Osterman, Driscoll, & Drake, 2018). Thus, the study of Latina mental health during the perinatal period is of high importance.

Taking a biopsychosocial conceptual approach to risk of postpartum depression, Lara-Cinisomo and colleagues (2016) outline two primary pathways that lead to adverse physiological and psychological outcomes for Latinas living in the United States. The contextual stress pathway considers the stressors that arise from the broader social environment (e.g., poverty, trauma) which have an effect on PPD symptoms regardless of ethno-racial membership. While the cultural stress pathway considers stressors common among Latinx individuals living in the United States because of their ethnicity and migration history, such as limited English proficiency and discrimination. Contextual and cultural pathways independently and interactively predict risk for postpartum depressive symptoms. In this article, we test only the contextual and cultural pathways (excluding biomarkers), in an effort to further refine the psychosocial predictors included in conceptual model in two ways. First, by examining the proposed predictors of cultural and contextual stress longitudinally in order to strengthen their causal links to postpartum depression. Second, by testing the effects of cultural stress, over and above contextual stress to better understand the relative effects of each pathway on postpartum depressive symptom trajectories.

Contextual Stress

Latinas in the United States are at heightened risk for living in poverty, and in 2015 about one in five Latinas lived in poverty as compared to 9.6% of White women (Tucker & Lowell, 2016). Low-income women of primarily Latinx decent living in the United States have shown rates of postpartum depression diagnoses two times higher than women of middle income (Yonkers et al., 2001). In a community sample of 106 low-income Dominican and Puerto Rican pregnant women, 43% reported mild to moderate depressive symptoms during their third trimester, and an additional 10% reported severe depressive symptoms (Zayas, Jankowski, & McKee, 2003), a striking finding given the strong associations between prenatal and postnatal depressive symptoms (Robertson, Grace, Wallington, & Stewart, 2004). Additionally, in study of 1,662 recent mothers, financial hardship was found to triple the risk for postpartum depressive symptoms and explained the greatest amount of variability in symptoms for Black and Latina women (Rich-Edwards et al., 2006). Finally, results from qualitative studies reveal that women living in poverty often discuss financial worries as the perceived cause of their depression during the postpartum period and view their poverty as a maternal failure (Abrams & Curran, 2011). These findings suggest that poverty is a robust contextual risk factor for postpartum depression in the general population, and a factor that may increase risk disproportionately in minority populations, given higher rates of financial hardship in these groups.

Domestic violence is another factor that is central to existing conceptual frameworks of postpartum depression in Latinas, considered a lifetime trauma (Lara-Cinisomo, Girdler, Grewen, & Meltzer-Brody, 2016). Previous research has focused on intimate partner violence (IPV; Beydoun, Beydoun, Kaufman, Lo, & Zonderman, 2012). Though rates of IPV are similar among Latinas and non-Latina white women (Breiding, Chen, & Black, 2014), Latinas who experience IPV tend to be younger, have more socioeconomic disadvantage than their non-Latina-white counterparts and be more fearful of seeking help from authorities (Bauer, Rodriguez, Quiroga, & Flores-Ortiz, 2000; Klevens, 2007; Lipsky, Caetano, & Roy-Byrne, 2009). IPV clearly contributes to depressive symptoms among Latinas irrespective of its temporal proximity to the pregnancy. For example, Latinas who experienced IPV in the year prior to their pregnancy report higher postpartum depressive symptoms (Valentine, Rodriguez, Lapeyrouse, & Zhang, 2011), as do Latinas who endorse experiences of IPV during their pregnancy (Jackson et al., 2015), and at any point in their past (Rodriguez et al., 2008; Rodríguez et al., 2010). However, there is also evidence that physical and sexual abuse perpetrated by a nonromantic partner (e.g., family member) prior to (Silverman & Loudon, 2010) or during a woman's pregnancy (Jasinski, 2004) are associated with depressive symptoms in the postpartum period. Because Latina mothers in the United States are more likely than other racialethnic groups to live in homes with multiple adult family members, examining domestic violence, which extends beyond violence perpetrated by an intimate partner is an important extension of the study of violence and postpartum depression in this group (Tamis-Lemonda & Kahana-Kalman, 2009).

Cultural Stress

Discrimination constitutes a cultural stressor commonly encountered by Latinas (Lee & Ahn, 2012) based on factors like language, accent, skin color, and immigrant status, and it contributes to poorer mental health outcomes in this group (Araújo & Borrell, 2006; Torres & Ong, 2010; Williams, Neighbors, & Jackson, 2003). Discrimination is reliably associated with greater risk for depression in ethno-racial minority women (Belle & Doucet, 2003; Canady, Bullen, Holzman, Broman, & Tian, 2008; Greer, 2011; Schulz et al., 2006), and heightens risk for delivering preterm birth (Fryer, Vines, & Stuebe, 2019; Mustillo et al., 2004) and low birth weight babies (Collins et al., 2004)-independent risk factors for postpartum depression (Bugental, Beaulieu, & Schwartz, 2008; Treyvaud, Lee, Doyle, & Anderson, 2014). Though relationships between discrimination and depression remain understudied in the perinatal period, research on African American women has shown that everyday racial discrimination during pregnancy and postpartum is associated with adverse mental health outcomes. In one large prospective study, African American women's experiences of interpersonal discrimination during the first trimester were associated with more depressive symptoms at midpregnancy (Ertel et al., 2012). This same pattern emerged in a sample of African American and Latina mothers studied postpartum; that is, women who reported more lifetime discrimination also reported more depressive symptoms in the year after giving birth (Surkan, Peterson, Hughes, & Gottlieb, 2006). Latinas and African American women encounter more instances of discrimination than white women due to intersectional experiences of discrimination, or gendered racism, which has been linked to increased stress during the perinatal period (Rosenthal & Lobel, 2018). Taken together, the literature on discrimination indicates that it is a prevalent stressor with a unique influence on adverse outcomes in the postpartum period.

Place of birth, or *nativity*, is a sociodemographic variable and a proxy for one form of cultural stress associated with Latinas' risk for postpartum depression. A recent meta-analysis found that immigrant women have 1.5 to 2 times greater risk of postpartum depression compared to women born in the United States. (Falah-Hassani, Shiri, Vigod, & Dennis, 2015). Bashiri and Spielvogel (1999) suggest that cultural differences surrounding birth and postpartum practices increase an immigrant woman's risk for postpartum depressive symptoms. These cultural differences are exacerbated by the fact that many immigrant women are separated from important sources of extended family support during pregnancy, birth and postpartum. However, effects of foreign-born status may depend on the age at which an individual immigrates to the United States. In a nationally representative sample of 2,554 Latinx in the United States, Alegría and colleagues (2007) reported that arrival to the United States at age 7 or younger among women was associated with a lower risk for depression as compared to women who arrived later in life, due in part to fewer barriers to upward mobility and English language acquisition.

Finally, although findings are mixed, language preference has also garnered attention as a risk factor for postpartum depression in Latinas. In some cases, English language preference is associated with an increased risk (Davila, McFall, & Cheng, 2009; Martinez-Schallmoser, Telleen, & MacMullen, 2003). However, other studies have reported that Latinas' language preference is unrelated to postpartum depression risk (Beck, 2006; Valentine et al., 2011; see also Heilemann, Frutos, Lee, & Kury, 2004). The discrepant findings regarding language preference may be better understood if we consider that language preference is often used as a proxy for acculturation. Acculturation is a construct that describes the dual process of cultural and psychological change that results from contact between two cultures (Berry, 2005). Though using language as a proxy measure of acculturation has been criticized for focusing on only one behavioral aspect of culture (Lopez-Class, Castro, & Ramirez, 2011), a psychometric evaluation of singleitem language measures show that they can in fact provide a brief and valid measure of acculturation (Mills, Malcarne, Fox, & Sadler, 2014).

Within the Latinx community, acculturation is simultaneously associated with adverse health behaviors and mental health outcomes during the perinatal period, and with increased health care utilization (Lara, Gamboa, Kahramanian, Morales, & Hayes Bautista, 2005). Thus, it is possible that Latinas who show a preference for English language are at a higher risk for postpartum depression because their greater acculturation has coincided with behavioral changes that contribute to the loss of privileged health status found in less acculturated women (i.e., the immigrant health paradox). However, Spanish speaking women, who may be less acculturated, can encounter difficulties navigating new perinatal systems of care (Shaffer, 2002) and may lack the social support available to them in their countries of origin (McGlade, Saha, & Dahlstrom, 2004). In sum, findings remain inconclusive as to whether language preference accounts for variability in PPD symptoms in Latinas and suggest a need to better measure acculturation in studies of perinatal mental health.

Present Study

The present study's aims were twofold. First, we sought to further document the frequency, severity, and types of discrimination encountered by Latina mothers, because this stressor is understudied in relation to the postpartum experiences of Latinas. Second, based on the hypothesized contextual and cultural stress pathways to postpartum depressive symptoms identified by Lara-Cinisomo et al. (2016), we examined whether poverty, exposure to domestic violence, discrimination, foreign-born status, English language preference, and age of immigration are associated with trajectories of postpartum depressive symptoms. Further, we tested whether cultural stressors would provide a better fitting model of symptom trajectories over and above a model considering only contextual stressors. These aims were tested in a community sample of low-income U.S.-born and foreign-born Latinas from various regions of the United States who were studied longitudinally in the year following a birth. We hypothesized that poverty and exposure to domestic violence (contextual stressors), as well as discrimination and foreign-born status (cultural stressors) would be associated with higher depressive symptoms at 1-month postpartum, and that these associations would persist over time, at 6 and 12 months postpartum. Due to the conflicting findings with regard to the effects of language (English/Spanish), and sparse findings on age of immigration in the context of postpartum depression, analyses on these variables were exploratory.

Method

The present study utilized data collected in 2003-2008 by the Community Child Health Network (CCHN), a multisite community-based participatory research collaboration with sites in Los Angeles, Chicago, Washington, DC, Baltimore, and eastern North Carolina (Ramey et al., 2015). CCHN study aims were to measure stress and resilience in mothers after birth to predict postpartum allostatic load and address maternal and child health disparities. Study eligibility criteria were that mothers (a) selfidentify as African American, Hispanic, or non-Hispanic White (b) be between 18 and 40 years of age; (c) be willing to complete interviews in either English or Spanish; (d) be residents in one of the identified communities for at least 6 months; (e) have 4 or fewer children including the new infant; and, (f) not have plans to be surgically sterilized following birth. Interested mothers were provided with study information, and informed consent was obtained for those who agreed to participate. Trained bilingual interviewers conducted face-to-face interviews in women's homes at three time points: approximately 1-month postpartum (T1), 6-months postpartum (T2), and 12-months postpartum (T3). Interviews included multiple measures of stress and resilience, physical and mental health, and sociodemographic information. The study protocol was approved by the institutional review board of each participating site.

547

Participants

In the present study, the sample consists of 537 mothers with complete data out of 607 total Latinas in the CCHN maternal cohort; 1,349 African American and 554 White women from the larger sample are excluded. Women in this sample lived primarily in Chicago (47%) and Los Angeles (30%) and were disproportionately very low-income with 42% of women reporting a household income at or below the federal poverty line and a mean per capita household income (adjusted for cost of living in study sites) of \$9,750.66 (SD = 13,306.84). A majority of the study sample was foreign born (71%), most were of Mexican descent (68%) and nearly 60% preferred to speak Spanish during the interviews. For additional demographic characteristics of the sample, see Table 1.

Measures

Poverty. The first interview included questions on household income and number of people living in the household. CCHN calculated three poverty categories based on the U.S. Census Bureau, Weighted Average Poverty Thresholds, 2009 (U.S. Census Bureau, 2009), which vary according to the size of the household: (a) $\leq 100\%$ federal poverty level (FPL; indicating income at or less than poverty threshold); (b) 101–200% FPL (indicating near poverty); and (c) $\geq 200\%$ FPL (indicating no poverty). Two dummy codes were used to represent the three groups, where $\leq 100\%$ FPL, or poverty, was the reference group.

Domestic violence. Domestic violence was measured using a modified version of Hurt, Insulted, Threatened with Harm and Screamed (HITS), a standard screen for domestic violence that was

| Table 1 | | |
|---------|------------------------|------|
| Sample | Characteristics $(n =$ | 537) |

| Sociodemographic variables | $\begin{array}{c} \text{Mean (SD) or} \\ n (\%) \end{array}$ | Range |
|------------------------------------|--|-----------|
| Age | 25.70 (5.25) | 18-40 |
| Number of prior births | | 0–4 |
| 0 (studied following first birth) | 216 (40.2) | |
| 1 (studied following second birth) | 218 (40.6) | |
| 2 (studied following third birth) | 99 (18.4) | |
| Nativity | | |
| U.S. born | 155 (28.9) | |
| Foreign born* | 382 (71.1) | |
| Spanish language preference | 318 (59.2) | |
| English language preference | 219 (40.8) | |
| Per capita household income | 9,750.66 (13,306.84) | 0-133,330 |
| Federal poverty level (FPL) | | |
| <100% FPL | 226 (42.1) | |
| >100% FPL | 220 (40.9) | |
| >200% FPL | 91 (17.0) | |
| Married to baby's father at T1 | 210 (39.1) | |
| Site | | |
| Los Angeles | 162 (30.2) | |
| Chicago | 284 (47.3) | |
| North Carolina | 4 (.6) | |
| Washington, D.C. | 88 (16.4) | |
| Education | | |
| Less than high school | 191 (35.6) | |
| Some college, no 4-year degree | 240 (44.6) | |
| College courses or college degree | 99 (18.4) | |

* Foreign born mothers where primarily from Mexico (67.8%) and El Salvador (11.0%).

administered to mothers at T1 (Sherin, Sinacore, Li, Zitter, & Shakil, 1998). The HITS includes four items related to physical hurt, insult, threats, and screaming toward self or others in the household and we included an additional item regarding domination or emotional control (O'Campo, Caughy, & Nettles, 2010). Mothers were asked whether they had encountered any violence item in the last year and specified whether the perpetrator was a "partner or spouse", "another family member" or "someone else in your household". Mothers responded using a 5-point frequency format (1 = never to 5 = frequently), with responses summed for a total score from 5 to 25. This scale has demonstrated acceptable internal and concurrent validity, good sensitivity and specificity (Rabin, Jennings, Campbell, & Bair-Merritt, 2009), and its use has been validated with Spanish speaking Latinas (Chen, Rovi, Vega, Jacobs, & Johnson, 2005).

Discrimination. Discrimination was measured using the Everyday Discrimination Scale (Williams, 1997) administered at T1 which is composed of 10 items that assess the general frequency of chronic and routine experiences of unfair treatment in an individual's day-to-day life. All items are scored on a six-point scale ranging from 0 = never to 5 = almost every day, such that higher scores indicate more frequent discrimination. For each discriminatory incident probed, women were able to attribute the experienced discrimination to one or more of the following reasons: (a) ancestry or national origins, (b) gender, (c) ethnicity/race, (d) shade of skin color, (e) language or accent, (f) sexual orientation, (g) age, or (h) height or weight. This scale has been validated in English and Spanish (Krieger, Smith, Naishadham, Hartman, & Barbeau, 2005), and has been used in community samples of immigrant Latinx (Cobb, Xie, Meca, & Schwartz, 2017). Cronbach's alpha coefficients in this sample were $\alpha = .86$ for Spanish speakers and $\alpha = .89$ for English speakers. Because Everyday Discrimination scores were positively skewed, a log transformation was conducted which resulted in a normal distribution.

Cultural demographic variables. All cultural demographic variables were assessed at T1. English versus Spanish language preference was determined based on the language that participants chose for the study interviews and was dummy coded (reference group = English language preference). Dummy variables were also created for U.S. versus foreign birth (reference group = U.S. birth). Participants reported the age at which they immigrated to the United States (if applicable). Age of immigration was divided into women who immigrated at age 7 or below (early arrival; Alegría, Sribney, Woo, Torres, & Guarnaccia, 2007), and those who immigrated after 7 years of age. Two dummy codes were used to represent U.S. born Latinas (vs. early arrival; reference group) and those who immigrated before versus after age 7 (reference group = early arrival).

Covariates. We included factors previously associated with postpartum depression in order to better determine the unique effect of the cultural and contextual stress variables on postpartum depressive symptoms. Three variables were identified as reliably linked to postpartum depression—negative life events (Boyd, Zayas, & McKee, 2006), marital status (Beck, 2001), and maternal education (Vliegen, Casalin, & Luyten, 2014). All covariates were measured at T1. The life events checklist was adapted by removing items inquiring about discrimination and domestic violence assessed by other multiitem measures and assessed whether a series of 24 stressful events (e.g., death of a loved one) occurred in the

year prior to the birth of the child. Adaptations were consistent with its use in prior studies (Dominguez, Schetter, Mancuso, Rini, & Hobel, 2005; Tanner Stapleton et al., 2016) which along with others show predictive validity for depression in the perinatal period (Falah-Hassani, Shiri, & Dennis, 2016). Any endorsed event was summed into a Life Events Count score. Women were also asked if they were married to the baby's father (reference group = unmarried) and to report on their educational attainment. For our analyses, women who pursued schooling beyond a high school degree were compared to a reference group who did not obtain additional schooling after high school or did not receive a high school degree.

Depressive symptoms. The Edinburgh Postnatal Depression Scale (EPDS; Cox, Holden, & Sagovsky, 1987) was administered at all three time points following birth. The EPDS is a measure used to screen individuals at risk for perinatal depression and it is validated in both English (e.g., Murray & Carothers, 1990) and Spanish (Alvarado, Jadresic, Guajardo, & Rojas, 2015; Garcia-Esteve, Ascaso, Ojuel, & Navarro, 2003). Further, the EPDS demonstrates cross ethnic/racial measurement equivalence (Di Florio et al., 2017). It is comprised of 10 items that assess the severity of common depressive symptoms experienced in the previous 7 days. All questions are scored on a Likert scale from 0 ("not at all") to 3 ("almost always/most of the time"), with higher scores indicating more depressive symptoms. Though analyses were conducted on the continuous measure, for descriptive purposes, women were categorized as having possible depression with EPDS scores between 10 and 12, and probable depression with scores >12. Cronbach's alpha coefficients for the scale across the three time points ranged from .79 to .83 for Spanish speakers and from .80 to .87 for English speakers.

Data Analysis Plan

Nested multilevel growth models were used to examine the effect of cultural stress (i.e., discrimination, language preference, foreign-born status, and age of immigration) over and above contextual stress (i.e., poverty and domestic violence) on the longitudinal growth trajectory of postpartum depressive symptoms. To handle missing data (29% attrition T2, and 22% at T3), models were estimated by a full information maximum likelihood (FIML), which provides unbiased parameter estimates and standard errors, particularly useful for longitudinal designs with data missing at random (Ferro, 2014). FIML was considered an appropriate way to handle missing data, as comparisons between the means of all variables included in the model did not differ systematically between participants who missed study visits versus those that did not. All analyses were conducted in HLM 7.03 (Raudenbush, Bryk, & Congdon, 2017).

The first step of the analysis was to determine the functional form of the trajectory of PPD symptoms over three time points, that is, one, six, and 12 months after birth. This was done by fitting two unconditional models, a linear and quadratic time model. If the quadratic term in the latter model was significant, that model would be used for the second step of the analysis; otherwise, the linear time model would be used. For both models, time was coded as a discrete variable, such that T1 was coded as zero, T2 as one, and T3 as two.

The second step of the analysis was to determine the best fitting model. To do this, three nested models were specified. The initial model was a covariates-only model that included negative life events, maternal education, and marital status at Level 2. The contextual stressors model added poverty and domestic violence variables to the model at Level 2. The final model, the combined stressors model, included all terms in the previous model and added cultural stress variables: discrimination, foreign-born status, early immigration, and language preference to Level 2. The multilevel equations for the linear time form of the final model were as follows, for each individual *i* at each time point *t*:

$$PPD_{ti} = \pi_{0i} + \pi_{1i} \times Time_{ti} + e_{ti}$$

 $\pi_{0i} = \beta_{00} + \beta_{01} \times Marital \; Status_i + \beta_{02} \times Maternal \; Education_i$

- + β_{03} × Negative Life Events_i + β_{04} × Near Poor_i
- + $\beta_{05} \times Not Poor_i + \beta_{06} \times Domestic Violence_i$
- + $\beta_{07} \times \text{Foreign Born}_i + \beta_{08} \times \text{Language}_i$
- + β_{09} × Late vs Early Immigration_i
- $$\label{eq:stars} \begin{split} &+ \beta_{010} \times US \text{ vs Early Immigration}_i + \beta_{011} \text{ Discrimination}_i \\ &+ r_{n_i} \end{split}$$

 $\pi_{11} = \beta_{10} + \beta_{11} \times \text{Marital Status}_i + \beta_{12} \times \text{Maternal Education}_i$

- + β_{13} × Negative Life Events_i + β_{14} × Near Poor_i
- $+ \beta_{15} \times Not Poor_i + \beta_{16} \times Domestic Violence_i$
- + $\beta_{17} \times \text{Foreign Born}_i + \beta_{18} \times \text{Language}_i$
- + β_{19} × Late vs Early Immigration_i
- + $\beta_{010} \times US$ vs Early Immigration_i
- + β_{110} × Discrimination_i + r_{0i}

Continuous variables (everyday discrimination, negative life events, exposure to domestic violence) were centered around each overall mean (i.e., grand mean centering), and marital status, maternal education, poverty, foreign-born status, early immigration and language were dummy coded (reference groups described in methods section). Covariates and all study variables were included as predictors of the intercept (i.e., baseline depressive symptoms) and as predictors of the slope-which tested the interactions between predictors and time on our outcome (i.e., change in depressive symptoms over time). The intercept and the slopes were allowed to vary randomly, and models were estimated using full maximum likelihood. The differences between these models were tested using likelihood ratio tests. We hypothesized that the model that included both contextual and cultural stressors would be the best overall model. The third step of the analysis was to examine the significance of the predictors individually in the best-fitting or final model.

Results

Descriptive Statistics on Depression and Discrimination

The mean scores on the EPDS at one, six and 12 months after birth were 4.81 (SD = 4.47), 5.20 (SD = 4.34) and 5.47 (SD = 4.38). At one month, 8.7% of EPDS scores were in the range of possible depression (scores between 10 and 12), while 7.1% were in the range of probable depression (scores >12). At six months, percentages went up to 9.0% and 9.3% respectively, and at 1-year post birth, 9.9% of women met criteria for possible depression and an additional 9.6% for probable depression. More than two thirds of the women reported experiencing discrimination at least "a few times a year" (68%), whereas nearly one third (32%) reported experiencing discrimination were: "People act as if they are better than you"; "You are treated with less respect than other people" and "People act as if you are not smart". Women's attributions for their discriminatory experiences are found in Table 2.

Bivariate Correlations and Mean Differences of Study Variables

Bivariate correlations revealed that depressive symptoms at all time points were positively associated with everyday discrimination at T1 (all p values > .05) and with domestic violence at T1 (all p values > .05). Correlations between the remaining variables of interest, can be found in Table 3. On average, scores on domestic violence and discrimination were low (domestic violence: M =6.27; discrimination: M = 8.93), though they ranged widely (domestic violence: SD = 2.43; discrimination: SD = 7.12). Independent samples t tests revealed no mean differences in depressive symptoms at any time point based on language preference, foreignborn status and early immigration to the United States. However, English language preference and U.S. birth were associated with higher mean scores on both everyday discrimination (language: t(604) = 4.04, p < .001; nativity: t(604) = 4.16, p < .001), and domestic violence (language: t(538) = 1.97, p = .049; nativity: t(604) = 2.17. p = .031.

Examining Contextual and Cultural Pathways to PPD Symptoms Over Time

Regarding the form of the PPD trajectory over time, the fixed quadratic effect was not significant. Thus, time was incorporated as a linear effect (p < .05) in all succeeding models. The first likelihood ratio test, comparing the covariates-only and the con-

| Table 2 | | | | |
|-------------|-------------|----------------|-------------|-----------|
| Attribution | of Evervdav | Discrimination | Experiences | (n = 413) |

| Attribution | Frequency | Percent (%) | | |
|-----------------------------|-----------|-------------|--|--|
| Race or ethnicity | 204 | 49.4 | | |
| Language or accent | 184 | 44.6 | | |
| Ancestry or national origin | 127 | 30.8 | | |
| Age | 97 | 23.5 | | |
| Gender | 91 | 22.0 | | |
| Shade of skin or skin color | 89 | 21.5 | | |
| Height or weight | 50 | 12.1 | | |
| Missing | 9 | 2.2 | | |
| Sexual orientation | 8 | 1.9 | | |

Note. Participants were able to make more than one attribution of discrimination, as such, the percentages do not add up to 100. The mean number of attributions made by participants was about 2 (M = 2.08, SD = 1.55).

textual stress models, showed that the inclusion domestic violence and poverty into the model significantly improved model fit $(\chi^2(6) = 53.12, p < .001)$ compared to a model with covariates only. The second likelihood ratio test, comparing contextual stress and combined stress models, showed that the inclusion of cultural stress variables (discrimination, foreign-born status, language preference, and age of immigration) significantly improved model fit $(\chi^2(8) = 32.54, p < .001)$ compared to a model with just contextual stressors and covariates.

In the final and best fitting model, domestic violence ($\beta = .39$, SE = .09, p = < .001, 95% CI [.22, .57]) and discrimination ($\beta =$.13, SE = .02, p = < .001, 95% CI [.09, .18]) were each associated with higher depressive symptoms at one month postpartum, over and above marital status, maternal education and negative life events. Language preference, foreign born status, and early immigration to the United States were not significant predictors of EPDS symptoms at one month postpartum (all p values > .05). None of the interactions between time and contextual stressors, or cultural stressors were significant (all p values > 0.5), indicating that these stressors did not change the linear slope of depressive symptoms over the year. Only negative life events, a covariate, interacted significantly with time ($\beta = -.05$, SE = .02, p = .003, 95% CI [-.08, -.02]), indicating that mothers who endorsed more negative life events than average showed linear decreases in EPDS scores over the year postpartum compared to those endorsing fewer negative life events than average. To see all fixed effects, random effects and variance components for each nested model, see Table 4.

Discussion

This study was based on a conceptual model of postpartum depression specific to Latinas living in the United States (Lara-Cinisomo et al., 2016). To our knowledge, this is the first study to simultaneously test cultural and contextual stressors together as predictors of depressive symptoms in Latinas following the birth of a child and over one year postpartum. We found that those who experienced higher levels of discrimination and domestic violence had more depressive symptoms at 1-month postpartum controlling for several well-known risk factors. Yet neither discrimination nor domestic violence was associated with the linear trajectory of depressive symptoms over the year postpartum, suggesting that in this community sample, these stressors are associated with higher baseline symptomology but not with *changes* in symptomology from one to six, and six to 12 months postpartum.

In this sample, about 16% of women had possible or probable depression at one month postpartum, 18% at six months, and 19.5% at one year. Other prevalence studies of depression in community samples of Latinas in the immediate postpartum (e.g., 1-6 weeks) report rates ranging from 27% using the EPDS (Malek, Connolly, & Knaus, 2001) as in the present study, to 40% in a largely undocumented sample using the Center for Epidemiologic Studies Depression Scale (Kuo et al., 2004). Thus, the present findings are in the lower range of those estimates. None-theless, the fact that 1 in 5 women have notable symptoms of depression at one year is concerning, and study findings underscore the persistence of symptoms over the year in a community sample.

| Measure | 1 | 2 | 3 | 4 | 5 | 6 | 7 | 8 | 9 |
|--|--------|--------|--------|------------|--------|------------|-----------|-----------|---|
| 1. PPD symptoms T1 | 1 | | | | | | | | |
| 2. PPD symptoms T2 | .474** | 1 | | | | | | | |
| 3. PPD symptoms T3 | .450** | .610** | 1 | | | | | | |
| 4. Everyday discrimination | .358** | .226** | .200** | 1 | | | | | |
| 5. Foreign born | .014 | 026 | .023 | 017^{*} | 1 | | | | |
| 6. Language preference (English vs. Spanish) | 002 | 041 | 011 | 080^{*} | .598** | 1 | | | |
| 7. Poverty category | 058 | 077 | 005 | 068 | 014 | 190^{**} | 1 | | |
| 8. Domestic violence | .372** | .263** | .247** | .373** | 093* | 085^{*} | 089^{*} | 1 | |
| 9. Early arrival to U.S. | .097 | .012 | .070 | 180^{**} | .915** | .685** | 049 | 101^{*} | 1 |

| Table 3 | | | | | | |
|----------------------|-----------|-------|-----------|-----|----------|--------|
| Correlations Between | Variables | Under | Study for | the | Complete | Sample |

p < .05. p < .01.

Our study is among the first to explore the association between discrimination and postpartum depressive symptoms in Latinas with the exception of one study that had some Latinas in a mixed sample (Surkan et al., 2006). Women who endorsed greater than average experiences of everyday discrimination reported more depressive symptoms one month postpartum. These Latinas indicated that their ethnicity, language or accent, or ancestry/national origin were the bases of reported discrimination. Although evi-

dence increasingly demonstrates significant associations between racial discrimination and postpartum depression in African American women (Ertel et al., 2012; Heldreth et al., 2016), this phenomenon has rarely been examined in Latinas. Our findings point to the relevance of discrimination for the mental health of Latina mothers. Of note, these data were collected between 2003 and 2008, and discrimination against the Latinx community is higher now. Irrespective of political party, U.S. citizens perceive an

Table 4

Fixed and Random Effects With Robust Standard Errors for the Growth Models of Postpartum Depressive Symptoms (n = 537)

| | Model 1 (Covariates) | | | Model 2 essors + Covariates) | Model 3 (Contextual stressors + Cultural stressors + Covariates) | |
|--|---|---|---|--|---|--|
| Variable | β (SE) | 95% CI [LL, UL] | β (SE) | 95% CI [LL, UL] | β (SE) | 95% CI [LL, UL] |
| Intercept predictors Intercept Marital status Maternal education Negative life events Near poverty Not poor Domestic violence Foreign born Language Late v. early immigration U.S. v. early immigration Discrimination | 5.07 (.23) 47 (.35) .03 (.42) .21 (.03) | [4.84, 5.30] [-1.16, .22] [79, .85] [.15, .27] | 5.27 (.29) 38 (.35) .24 (.46) .16 (.03) 56 (.36) 22 (.55) .49 (.09) | [4.71, 5.84] [-1.07, .30] [66, 1.13] [.11, .22] [-1.27, .14] [-1.31, .86] [.32, .66] | 6.58 (1.62) 67 (.34) .32 (.45) .14 (.03) 49 (.35) .24 (.55) .39 (.09) -1.26 (1.56) .60 (.48) 21 (.74) -2.15 (1.64) .13 (.02) | [3.40, 9.76] [-1.35, 0] [57, 1.21] [.08, .19] [-1.39, .41] [85, 1.33] [.22, .57] [-4.31, 1.79] [35, 1.55] [-1.165, 1.23] [-5.37, 1.06] [.09, .18] |
| Slope predictors Intercept Marital status Maternal education Negative life events Near poverty Not poor Domestic violence Foreign born Language Late v. early immigration U.S. v. early immigration Discrimination | .38 (.16) .19 (.23) 58 (.27) 06 (.02) | [.07, .69] [26, .64] [-1.11,05] [10,02] | .27 (.21) .17 (.23) 58 (.33) 05 (.02) .24 (.25) .11 (.41) 04 (.05) | [14, .67] [29, .63] [-1.23, .07] [03,08] [24, .74] [71, .93] [14, .06] | $\begin{array}{c} .22 \ (1.49) \\ .27 \ (.23) \\61 \ (.32) \\05 \ (.02) \\ .25 \ (.25) \\02 \ (.42) \\01 \ (.05) \\13 \ (1.4) \\14 \ (.32) \\ .17 \ (.48) \\ .37 \ (1.49) \\03 \ (.02) \end{array}$ | $\begin{bmatrix} -2.71, 3.15 \\ [18, .72] \\ [-1.24, .01] \\ [08,02] \\ [25, .75] \\ [85, .81] \\ [12, .09] \\ [-2.96, 2.70] \\ [78, .49] \\ [78, 1.12] \\ [-2.54, 3.29] \\ [08, .00] \end{bmatrix}$ |
| Variance components Intercept (τ00) Time slope (τ11) L1 residual (σ2) Deviance | 8.37 1.05 8.52 6869.93 | | 6.90 .99 8.51 6797.72 | | 6.03 .97 8.48 6763.59 | |

Note. Bold coefficients and variance components are significant at p < .05.

increase in discrimination against Latinx since the last presidential election (Pew Research Center, 2019). Thus, future studies of perinatal health in Latinas should measure discrimination to determine rates in pregnancy and impact on birth and postpartum outcomes.

The finding that domestic violence experienced during pregnancy and early postpartum is associated with higher levels of postpartum depressive symptoms is consistent with prior research. Our measure of domestic violence included violence by an intimate partner, which has been identified as a risk factor of particular importance for postpartum depression in Latinas (Blackmore & Chaudron, 2014; Rodríguez et al., 2010). We expand upon past findings on violence during pregnancy and postpartum depressive symptoms by measuring violence perpetrated by anyone in the household. In our sample, about 45% of these Latinas reported experiencing violence in the home, and of those, three-quarters were reported to be by an intimate partner and one fourth by someone other than an intimate partner. These rates are well within the range of prevalence rates (4%-80%) of interpersonal violence calculated for Latinas in a recent systematic review (Gonzalez, Benuto, & Casas, 2018). Women who experience violence at the hands of nonintimate partners are at increased risk of also experiencing violence at the hands of intimate partners (Krebs, Breiding, Browne, & Warner, 2011), making this group of women particularly vulnerable to adverse mental health sequalae including depression and anxiety (Pico-Alfonso et al., 2006). Though not measured in our study, Latinas are more likely than women in other ethnic groups to live with extended family members during the perinatal period (Tamis-Lemonda & Kahana-Kalman, 2009). Future work should consider the effects of exposure to violence both in and beyond the context of an intimate relationship.

Contrary to our hypotheses, foreign-born status and poverty were not associated with trajectories of postpartum depressive symptoms. Regarding foreign-born status, previous studies that find no association between Latina women's country of origin and postpartum depression (Chaudron et al., 2005; Kuo et al., 2004) suggest that it is financial hardship that accounts for elevated risk in immigrant women (Rich-Edwards et al., 2006). In our sample, poverty was not significantly associated with depressive symptoms and virtually all the women in the sample were very low income (92.1% making \$25,000 or less/year). Thus, we likely had insufficient variability to replicate effects found in samples with a wider distribution on socioeconomic status (Goyal, Gay, & Lee, 2010; O'Hara & McCabe, 2013).

In addition, we found that mothers who endorsed more negative life events than average in the year preceding the birth of their baby showed linear decreases in symptoms over the year postpartum. This finding was unexpected, as previous studies have reported that a greater number of prenatal life events are associated with postpartum depressive symptoms in Latinas (e.g., Boyd et al., 2006; Luecken, Crnic, Gonzales, Winstone, & Somers, 2019). However, some evidence suggests that negative life events only lead to clinically significant depressive symptoms in women who endorse prior depression (Marks, Wieck, Checkley, & Kumar, 1991). This may point to biased cognitive processing of negative life events (e.g., difficulties disengaging from negative memories), and not the *number* of events as the underlying mechanism that links life events and depression (Gotlib & Joormann, 2010). The timing of the negative life events may also be a factor. At 1-month postpartum, women who reported an above average number of negative life events in the last year had more depressive symptoms, when events (e.g., a robbery, the death of a loved one) were most proximal. This is consistent with evidence that acute negative life events have their greatest impact on depressive symptoms in the short term (Brown & Rosellini, 2011), and that their effect reduces over time (e.g., Howe, Hornberger, Weihs, Moreno, & Neiderhiser, 2012).

Regarding limitations, although we adjusted for several variables previously associated with postpartum depression, we could not control for preexisting anxiety or depressive symptoms. Given the associations between prenatal and postnatal psychopathology (Milgrom et al., 2008), controlling for depressive symptomology during pregnancy in future studies can refine the estimate of the relative impact of cultural and contextual stressors on depressive symptoms during this important time in women's lives. Though we examined most variables in Lara-Cinisomo and colleagues (2016) model, we were not able to examine acculturative stress or border crossing trauma, as these variables were not assessed as part of the original study. Study variables that were included, such as discrimination, nativity and language preference, are likely to be associated with acculturative stress, since Latinx individuals who are foreign-born, prefer to speak Spanish, and report discrimination also score higher on measures of acculturative stress (Araújo Dawson & Panchanadeswaran, 2010; Bravo, Umaña-Taylor, Toomey, Updegraff, & Jahromi, 2016; Castillo et al., 2015; Torres, Driscoll, & Voell, 2012). Nonetheless, the aforementioned cultural stressors warrant future study, especially in longitudinal contexts.

While we intentionally limited our analyses to psychosocial factors, future work should examine biological mediators (i.e., markers of dysregulated HPA axis and oxytocin) to better understand perinatal physiological dysregulation and its mental health sequelae in Latinas. It should also be noted that women in this sample were primarily of Mexican decent, and these findings may not be generalizable to women from other places of origin such as Cuba, South or Central America. Nationally representative data from Latinx adults have shown differences in English language proficiency, migration experiences and family conflict based on an individual's county of origin—all important correlates of mental health outcomes (Guarnaccia et al., 2007). It is possible then, that the cultural and contextual risks we studied may occur at different rates or have differential effects on the postpartum mental health of Latinas of other origins residing in the United States.

In sum, we examined conceptually driven predictors of cultural and contextual stress in a sample of low-income U.S. and foreignborn Latinas. We identified discrimination and domestic violence as stressors that predicted symptom trajectories. Women who reported greater than average experiences of discrimination and domestic violence started with higher levels of depressive symptoms at one month postpartum. The lack of linear change in depressive symptoms associated with these stressors demonstrated that the impact of above average levels of discrimination and domestic violence persisted across the year. However, given that these cultural and contextual stressors were concurrently associated with depressive symptoms, we must consider the possibility of a bidirectional relationship between the stressors and depression. For example, in line with the integrative interpersonal frameworks for depression (Hames, Hagan, & Joiner, 2013), it is possible that women with higher baseline depressive symptoms hold PONTING ET AL.

core beliefs about rejection that make them more likely to perceive discrimination. It is also known that depression and interpersonal conflict are dynamic and intercorrelated (Coyne, 1976). Still, these longitudinal findings add to a conceptual model of risk for postpartum depression for Latinas living in the United States by demonstrating that stressors particular to Latinas' ethnic experience are important over and above contextual stressors that Latinas experience at disproportionate rates but are not culturally specific.

The extensive contact with health care systems during pregnancy provides a window of opportunity to employ brief interventions for women encountering violence in the home (World Health Organization, 2015). Depression interventions during the perinatal period with components to address domestic violence have had limited success (Jahanfar, Howard, & Medley, 2014), but their effectiveness and design warrant continued investigation given the high rate of domestic violence in our sample. Further, our findings support the inclusion of discrimination as a treatment target for precision cultural tailoring. This is a focused approach to intervention adaptation that identifies ways in which cultural variables (e.g., immigration related stressors) may be assessed at the individual level to activate intervention components to improve treatment fit. Though clinicians cannot intervene directly on discriminatory experiences, they can utilize various cognitive and acceptance based coping strategies as well as explore their client's ethnic identity as a way of reducing the effects of this cultural stressor on depression in the postpartum period. For example, active coping strategies for responding to personal discrimination (e.g., confrontation, seeking social support) are theorized to moderate the link between discrimination and heightened physiological stress responses, and in turn improve effects on mental health outcomes (Pascoe & Smart Richman, 2009). Thus, including content about coping with discrimination within interventions for postpartum depression may increase their effectiveness for ethnoracial minority women.

The field remains disproportionately focused on pathways of risk for postpartum depression in Latinas. Continued identification of factors that mitigate cultural and contextual risk to inform intervention work (Heilemann et al., 2004) during pregnancy when addressing stressors may prevent postpartum depression. Moreover, Latinas remain an understudied group in research on perinatal mood disorders with limited evidence-based treatments tested for use in this population (Zayas & Sampson, 2014), and few interventions available in Spanish (Lara-Cinisomo & Wisner, 2014), making translational research on cultural and contextual risk and resilience a priority. Findings from this study set the stage for future translational work examining both cultural and contextual determinants of postpartum mental health in Latinas, who remain an underserved group.

References

- Abrams, L. S., & Curran, L. (2011). Maternal identity negotiations among low-income women with symptoms of postpartum depression. *Qualitative Health Research*, 21, 373–385. http://dx.doi.org/10.1177/ 1049732310385123
- Alegría, M., Sribney, W., Woo, M., Torres, M., & Guarnaccia, P. (2007). Looking beyond nativity: The relation of age of immigration, length of residence, and birth cohorts to the risk of onset of psychiatric disorders for Latinos. *Research in Human Development*, *4*, 19–47. http://dx.doi .org/10.1080/15427600701480980

- Alvarado, R., Jadresic, E., Guajardo, V., & Rojas, G. (2015). First validation of a Spanish-translated version of the Edinburgh postnatal depression scale (EPDS) for use in pregnant women: A Chilean study. Archives of Women's Mental Health, 18, 607–612. http://dx.doi.org/10.1007/ s00737-014-0466-z
- Araújo, B. Y., & Borrell, L. N. (2006). Understanding the link between discrimination, mental health outcomes, and life chances among Latinos. *Hispanic Journal of Behavioral Sciences*, 28, 245–266. http://dx.doi.org/ 10.1177/0739986305285825
- Bashiri, N., & Spielvogel, A. M. (1999). Postpartum depression: A crosscultural perspective. *Primary Care Update for Ob/Gyns*, 6, 82–87. http://dx.doi.org/10.1016/S1068-607X(99)00003-7
- Bauer, H. M., Rodriguez, M. A., Quiroga, S. S., & Flores-Ortiz, Y. G. (2000). Barriers to health care for abused Latina and Asian immigrant women. *Journal of Health Care for the Poor and Underserved*, 11, 33–44. http://dx.doi.org/10.1353/hpu.2010.0590
- Beck, C. T. (2001). Predictors of postpartum depression: An update. Nursing Research, 50, 275–285. http://dx.doi.org/10.1097/00006199-200109000-00004
- Beck, C. T. (2006). Acculturation: Implications for perinatal research. *The American Journal of Maternal/Child Nursing*, 31, 114–120. http://dx .doi.org/10.1097/00005721-200603000-00011
- Belle, D., & Doucet, J. (2003). Poverty, inequality, and discrimination as sources of depression among U.S. women. *Psychology of Women Quarterly*, 27, 101–113. http://dx.doi.org/10.1111/1471-6402.00090
- Berry, J. W. (2005). Acculturation: Living successfully in two cultures. International Journal of Intercultural Relations, 29, 697–712. http://dx .doi.org/10.1016/j.ijintrel.2005.07.013
- Beydoun, H. A., Beydoun, M. A., Kaufman, J. S., Lo, B., & Zonderman, A. B. (2012). Intimate partner violence against adult women and its association with major depressive disorder, depressive symptoms and postpartum depression: A systematic review and meta-analysis. *Social Science & Medicine*, 75, 959–975. http://dx.doi.org/10.1016/j .socscimed.2012.04.025
- Blackmore, E. R., & Chaudron, L. (2014). Psychosocial and cultural considerations in detecting and treating depression in Latina perinatal women in the United States. In S. Lara-Cinisomo & K. L. Wisner (Eds.), *Perinatal depression among Spanish-speaking and Latin American women* (pp. 83–96). New York, NY: Springer. http://dx.doi.org/10.1007/ 978-1-4614-8045-7 6
- Boyd, R. C., Zayas, L. H., & McKee, M. D. (2006). Mother-infant interaction, life events and prenatal and postpartum depressive symptoms among urban minority women in primary care. *Maternal and Child Health Journal*, 10, 139–148. http://dx.doi.org/10.1007/s10995-005-0042-2
- Bravo, D. Y., Umaña-Taylor, A. J., Toomey, R. B., Updegraff, K. A., & Jahromi, L. B. (2016). Risky behaviors and educational attainment among young Mexican-origin mothers: The role of acculturative stress and the educational aspiration-expectation gap. *International Journal of Intercultural Relations*, 52, 13–26. http://dx.doi.org/10.1016/j.ijintrel .2016.02.003
- Breiding, M. J., Chen, J., & Black, M. C. (2014). Intimate partner violence in the United States–2010. Atlanta, GA: National Center for Injury Prevention and Control, Centers for Disease Control and Prevention.
- Bromberger, J. T., Harlow, S., Avis, N., Kravitz, H. M., & Cordal, A. (2004). Racial/ethnic differences in the prevalence of depressive symptoms among middle-aged women: The Study of Women's Health Across the Nation (SWAN). *American Journal of Public Health*, 94, 1378– 1385. http://dx.doi.org/10.2105/ajph.94.8.1378
- Brown, T. A., & Rosellini, A. J. (2011). The direct and interactive effects of neuroticism and life stress on the severity and longitudinal course of depressive symptoms. *Journal of Abnormal Psychology*, *120*, 844–856. http://dx.doi.org/10.1037/a0023035

- Brummelte, S., & Galea, L. A. (2016). Postpartum depression: Etiology, treatment and consequences for maternal care. *Hormones and Behavior*, 77, 153–166. http://dx.doi.org/10.1016/j.yhbeh.2015.08.008
- Bugental, D. B., Beaulieu, D., & Schwartz, A. (2008). Hormonal sensitivity of preterm versus full-term infants to the effects of maternal depression. *Infant Behavior and Development*, 31, 51–61. http://dx.doi.org/10 .1016/j.infbeh.2007.06.003
- Canady, R. B., Bullen, B. L., Holzman, C., Broman, C., & Tian, Y. (2008). Discrimination and symptoms of depression in pregnancy among African American and White women. *Women's Health Issues*, 18, 292–300. http://dx.doi.org/10.1016/j.whi.2008.04.003
- Castillo, L. G., Cano, M. A., Yoon, M., Jung, E., Brown, E. J., Zamboanga, B. L., . . . Whitbourne, S. K. (2015). Factor structure and factorial invariance of the Multidimensional Acculturative Stress Inventory. *Psychological Assessment*, 27, 915–924. http://dx.doi.org/10.1037/ pas0000095
- Chang, J. J., Tabet, M., Elder, K., Kiel, D. W., & Flick, L. H. (2016). Racial/ethnic differences in the correlates of mental health services use among pregnant women with depressive symptoms. *Maternal and Child Health Journal*, 20, 1911–1922. http://dx.doi.org/10.1007/s10995-016-2005-1
- Chaudron, L. H., Kitzman, H. J., Peifer, K. L., Morrow, S., Perez, L. M., & Newman, M. C. (2005). Prevalence of maternal depressive symptoms in low-income Hispanic women. *The Journal of Clinical Psychiatry*, 66, 418–423. http://dx.doi.org/10.4088/JCP.v66n0402
- Chen, P.-H., Rovi, S., Vega, M., Jacobs, A., & Johnson, M. S. (2005). Screening for domestic violence in a predominantly Hispanic clinical setting. *Family Practice*, 22, 617–623. http://dx.doi.org/10.1093/ fampra/cmi075
- Cobb, C. L., Xie, D., Meca, A., & Schwartz, S. J. (2017). Acculturation, discrimination, and depression among unauthorized Latinos/as in the United States. *Cultural Diversity and Ethnic Minority Psychology*, 23, 258–268. http://dx.doi.org/10.1037/cdp0000118
- Collins, J. W., Jr., David, R. J., Handler, A., Wall, S., & Andes, S. (2004). Very low birthweight in African American infants: The role of maternal exposure to interpersonal racial discrimination. *American Journal of Public Health*, 94, 2132–2138. http://dx.doi.org/10.2105/ajph.94.12 .2132
- Cox, J. L., Holden, J. M., & Sagovsky, R. (1987). Detection of postnatal depression. Development of the 10-item Edinburgh Postnatal Depression Scale. *The British Journal of Psychiatry*, 150, 782–786. http://dx .doi.org/10.1192/bjp.150.6.782
- Coyne, J. C. (1976). Toward an interactional description of depression. *Psychiatry*, *39*, 28–40. http://dx.doi.org/10.1080/00332747.1976 .11023874
- Davila, M., McFall, S. L., & Cheng, D. (2009). Acculturation and depressive symptoms among pregnant and postpartum Latinas. *Maternal and Child Health Journal*, 13, 318–325. http://dx.doi.org/10.1007/s10995-008-0385-6
- Dawson, B. A., & Panchanadeswaran, S. (2010). Discrimination and acculturative stress among first-generation Dominicans. *Hispanic Journal* of Behavioral Sciences, 32, 216–231. http://dx.doi.org/10.1177/ 0739986310364750
- Di Florio, A., Putnam, K., Altemus, M., Apter, G., Bergink, V., Bilszta, J., ... Meltzer-Brody, S. (2017). The impact of education, country, race and ethnicity on the self-report of postpartum depression using the Edinburgh Postnatal Depression Scale. *Psychological Medicine*, 47, 787– 799. http://dx.doi.org/10.1017/S0033291716002087
- Dominguez, T. P., Schetter, C. D., Mancuso, R., Rini, C. M., & Hobel, C. (2005). Stress in African American pregnancies: Testing the roles of various stress concepts in prediction of birth outcomes. *Annals of Behavioral Medicine*, 29, 12–21. http://dx.doi.org/10.1207/s15324796 abm2901_3

- Ertel, K. A., James-Todd, T., Kleinman, K., Krieger, N., Gillman, M., Wright, R., & Rich-Edwards, J. (2012). Racial discrimination, response to unfair treatment, and depressive symptoms among pregnant black and African American women in the United States. *Annals of Epidemiology*, 22, 840–846. http://dx.doi.org/10.1016/j.annepidem.2012.10.001
- Falah-Hassani, K., Shiri, R., & Dennis, C. L. (2016). Prevalence and risk factors for comorbid postpartum depressive symptomatology and anxiety. *Journal of Affective Disorders*, 198, 142–147. http://dx.doi.org/10 .1016/j.jad.2016.03.010
- Falah-Hassani, K., Shiri, R., Vigod, S., & Dennis, C. L. (2015). Prevalence of postpartum depression among immigrant women: A systematic review and meta-analysis. *Journal of Psychiatric Research*, 70, 67–82. http://dx.doi.org/10.1016/j.jpsychires.2015.08.010
- Ferro, M. A. (2014). Missing data in longitudinal studies: Cross-sectional multiple imputation provides similar estimates to full-information maximum likelihood. *Annals of Epidemiology*, 24, 75–77. http://dx.doi.org/ 10.1016/j.annepidem.2013.10.007
- Field, T. (2010). Postpartum depression effects on early interactions, parenting, and safety practices: A review. *Infant Behavior & Development*, 33, 1–6. http://dx.doi.org/10.1016/j.infbeh.2009.10.005
- Fryer, K. E., Vines, A. I., & Stuebe, A. M. (2019). A Multisite examination of everyday discrimination and the prevalence of spontaneous preterm birth in African American and Latina women in the United States. *American Journal of Perinatology*. Advance online publication. http:// dx.doi.org/10.1055/s-0039-1693696
- Fung, K., & Dennis, C. L. (2010). Postpartum depression among immigrant women. *Current Opinion in Psychiatry*, 23, 342–348. http://dx.doi.org/ 10.1097/YCO.0b013e32833ad721
- Garcia-Esteve, L., Ascaso, C., Ojuel, J., & Navarro, P. (2003). Validation of the Edinburgh postnatal depression scale (EPDS) in Spanish mothers. *Journal of Affective Disorders*, 75, 71–76. http://dx.doi.org/10.1016/ S0165-0327(02)00020-4
- Gavin, N. I., Gaynes, B. N., Lohr, K. N., Meltzer-Brody, S., Gartlehner, G., & Swinson, T. (2005). Perinatal depression: A systematic review of prevalence and incidence. *Obstetrics and Gynecology*, *106*, 1071–1083. http://dx.doi.org/10.1097/01.AOG.0000183597.31630.db
- Gonzalez, F. R., Benuto, L. T., & Casas, J. B. (2018). Prevalence of interpersonal violence among Latinas: A systematic review. *Trauma*, *Violence & Abuse*. Advance online publication. http://dx.doi.org/10 .1177/1524838018806507
- Gotlib, I. H., & Joormann, J. (2010). Cognition and depression: Current status and future directions. *Annual Review of Clinical Psychology*, 6, 285–312. http://dx.doi.org/10.1146/annurev.clinpsy.121208.131305
- Goyal, D., Gay, C., & Lee, K. A. (2010). How much does low socioeconomic status increase the risk of prenatal and postpartum depressive symptoms in first-time mothers? *Women's Health Issues*, 20, 96–104. http://dx.doi.org/10.1016/j.whi.2009.11.003
- Greer, T. M. (2011). Coping strategies as moderators of the relationship between race-and gender-based discrimination and psychological symptoms for African American women. *Journal of Black Psychology*, 37, 42–54. http://dx.doi.org/10.1177/0095798410380202
- Guarnaccia, P. J., Martinez Pincay, I., Alegría, M., Shrout, P., Lewis-Fernández, R., & Canino, G. (2007). Assessing diversity among Latinos: Results from the NLAAS. *Hispanic Journal of Behavioral Sciences*, 29, 510–534. http://dx.doi.org/10.1177/0739986307308110
- Guo, N., Robakis, T., Miller, C., & Butwick, A. (2018). Prevalence of depression among women of reproductive age in the United States. *Obstetrics and Gynecology*, 131, 671–679. http://dx.doi.org/10.1097/ aog.000000000002535
- Hames, J. L., Hagan, C. R., & Joiner, T. E. (2013). Interpersonal processes in depression. *Annual Review of Clinical Psychology*, 9, 355–377. http:// dx.doi.org/10.1146/annurev-clinpsy-050212-185553
- Heilemann, M., Frutos, L., Lee, K., & Kury, F. S. (2004). Protective strength factors, resources, and risks in relation to depressive symptoms

among childbearing women of Mexican descent. *Health Care for Women International*, 25, 88-106. http://dx.doi.org/10.1080/07399330490253265

- Heldreth, C. M., Guardino, C. M., Wong, L. H., Schetter, C. D., Shapiro, J. R., Schafer, P., . . . Raju, T. (2016). Childhood racism experiences and postpartum depressive symptoms in African American mothers. *Journal* of Social and Clinical Psychology, 35, 840–864. http://dx.doi.org/10 .1521/jscp.2016.35.10.840
- Howe, G. W., Hornberger, A. P., Weihs, K., Moreno, F., & Neiderhiser, J. M. (2012). Higher-order structure in the trajectories of depression and anxiety following sudden involuntary unemployment. *Journal of Abnor*mal Psychology, 121, 325–338. http://dx.doi.org/10.1037/a0026243
- Howell, E. A., Mora, P. A., Horowitz, C. R., & Leventhal, H. (2005). Racial and ethnic differences in factors associated with early postpartum depressive symptoms. *Obstetrics and Gynecology*, *105*, 1442–1450. http://dx.doi.org/10.1097/01.AOG.0000164050.34126.37
- Jackson, C. L., Ciciolla, L., Crnic, K. A., Luecken, L. J., Gonzales, N. A., & Coonrod, D. V. (2015). Intimate partner violence before and during pregnancy: Related demographic and psychosocial factors and postpartum depressive symptoms among Mexican American women. *Journal of Interpersonal Violence*, 30, 659–679. http://dx.doi.org/10.1177/ 0886260514535262
- Jahanfar, S., Howard, L. M., & Medley, N. (2014). Interventions for preventing or reducing domestic violence against pregnant women. *Cochrane Database of Systematic Reviews*. Advance online publication. http://dx.doi.org/10.1002/14651858.CD009414.pub3
- Jasinski, J. L. (2004). Pregnancy and domestic violence: A review of the literature. *Trauma, Violence, & Abuse, 5,* 47–64. http://dx.doi.org/10 .1177/1524838003259322
- Klevens, J. (2007). An overview of intimate partner violence among Latinos. Violence Against Women, 13, 111–122. http://dx.doi.org/10 .1177/1077801206296979
- Kozhimannil, K. B., Trinacty, C. M., Busch, A. B., Huskamp, H. A., & Adams, A. S. (2011). Racial and ethnic disparities in postpartum depression care among low-income women. *Psychiatric Services*, 62, 619–625. http://dx.doi.org/10.1176/ps.62.6.pss6206_0619
- Krebs, C., Breiding, M. J., Browne, A., & Warner, T. (2011). The Association Between Different Types of Intimate Partner Violence Experienced by Women. *Journal of Family Violence*, 26, 487–500. http://dx .doi.org/10.1007/s10896-011-9383-3
- Krieger, N., Smith, K., Naishadham, D., Hartman, C., & Barbeau, E. M. (2005). Experiences of discrimination: Validity and reliability of a self-report measure for population health research on racism and health. *Social Science & Medicine*, *61*, 1576–1596. http://dx.doi.org/10.1016/j .socscimed.2005.03.006
- Kuo, W. H., Wilson, T. E., Holman, S., Fuentes-Afflick, E., O'Sullivan, M. J., & Minkoff, H. (2004). Depressive symptoms in the immediate postpartum period among Hispanic women in three U.S. cities. *Journal* of *Immigrant Health*, 6, 145–153. http://dx.doi.org/10.1023/B:JOIH .0000045252.10412.fa
- Lara, M., Gamboa, C., Kahramanian, M. I., Morales, L. S., & Hayes Bautista, D. E. (2005). Acculturation and Latino health in the United States: A review of the literature and its sociopolitical context. *Annual Review of Public Health*, 26, 367–397. http://dx.doi.org/10.1146/ annurev.publhealth.26.021304.144615
- Lara-Cinisomo, A., & Wisner, K. L. (Eds.). (2014). Perinatal depression among Spanish-speaking and Latin American women. New York, NY: Springer. http://dx.doi.org/10.1007/978-1-4614-8045-7
- Lara-Cinisomo, S., Girdler, S. S., Grewen, K., & Meltzer-Brody, S. (2016). A biopsychosocial conceptual framework of postpartum depression risk in immigrant and U.S.-born Latina mothers in the United States. Women's Health Issues, 26, 336–343. http://dx.doi.org/10.1016/j.whi.2016 .02.006

- Lee, D. L., & Ahn, S. (2012). Discrimination against Latina/os: A metaanalysis of individual-level resources and outcomes. *The Counseling Psychologist*, 40, 28–65. http://dx.doi.org/10.1177/0011000011403326
- Lipsky, S., Caetano, R., & Roy-Byrne, P. (2009). Racial and ethnic disparities in police-reported intimate partner violence and risk of hospitalization among women. *Women's Health Issues*, 19, 109–118. http:// dx.doi.org/10.1016/j.whi.2008.09.005
- Liu, C. H., & Tronick, E. (2012). Do patient characteristics, prenatal care setting, and method of payment matter when it comes to provider-patient conversations on perinatal mood? *Maternal and Child Health Journal*, *16*, 1102–1112. http://dx.doi.org/10.1007/s10995-011-0835-4
- Liu, C. H., & Tronick, E. (2013). Rates and predictors of postpartum depression by race and ethnicity: Results from the 2004 to 2007 New York City PRAMS survey (Pregnancy Risk Assessment Monitoring System). *Maternal and Child Health Journal*, *17*, 1599–1610. http://dx .doi.org/10.1007/s10995-012-1171-z
- Lopez-Class, M., Castro, F. G., & Ramirez, A. G. (2011). Conceptions of acculturation: A review and statement of critical issues. *Social Science & Medicine*, 72, 1555–1562. http://dx.doi.org/10.1016/j.socscimed.2011 .03.011
- Lucero, N. B., Beckstrand, R. L., Callister, L. C., & Sanchez Birkhead, A. C. (2012). Prevalence of postpartum depression among Hispanic immigrant women. *Journal of the American Academy of Nurse Practitioners*, 24, 726–734. http://dx.doi.org/10.1111/j.1745-7599.2012 .00744.x
- Luecken, L. J., Crnic, K. A., Gonzales, N. A., Winstone, L. K., & Somers, J. A. (2019). Mother-infant dyadic dysregulation and postpartum depressive symptoms in low-income Mexican-origin women. *Biological Psychology*, 147, 107614.
- Malek, L. L., Connolly, T. P., & Knaus, J. V. (2001). Cultural aspects of postpartum depressive symptoms: An urban Hispanic population. *Jour*nal of Women's Health, 1, 273–278.
- Marks, M. N., Wieck, A., Checkley, S. A., & Kumar, R. (1991). Life stress and post-partum psychosis: A preliminary report. *The British Journal of Psychiatry*, 158(Suppl. 10), 45–49. http://dx.doi.org/10.1192/ s0007125000291988
- Martin, J. A., Hamilton, B. E., Osterman, M. J. K., Driscoll, A. K., & Drake, P. (2018). Births: Final data for 2016. *National Vital Statistics Reports*, 67, 1–55.
- Martinez-Schallmoser, L., Telleen, S., & MacMullen, N. J. (2003). The effect of social support and acculturation on postpartum depression in Mexican American women. *Journal of Transcultural Nursing*, 14, 329– 338. http://dx.doi.org/10.1177/1043659603257162
- McGlade, M. S., Saha, S., & Dahlstrom, M. E. (2004). The Latina paradox: An opportunity for restructuring prenatal care delivery. *American Journal of Public Health*, 94, 2062–2065. http://dx.doi.org/10.2105/AJPH.94 .12.2062
- Milgrom, J., Gemmill, A. W., Bilszta, J. L., Hayes, B., Barnett, B., Brooks, J., . . . Buist, A. (2008). Antenatal risk factors for postnatal depression: A large prospective study. *Journal of Affective Disorders*, *108*, 147–157. http://dx.doi.org/10.1016/j.jad.2007.10.014
- Mills, S. D., Malcarne, V. L., Fox, R. S., & Sadler, G. R. (2014). Psychometric Evaluation of the Brief Acculturation Scale for Hispanics. *Hispanic Journal of Behavioral Sciences*, 36, 164–174. http://dx.doi .org/10.1177/0739986314526697
- Muñoz, R. F., Le, H. N., Ippen, C. G., Diaz, M. A., Urizar, G. G., Jr., Soto, J., . . . Lieberman, A. F. (2007). Prevention of postpartum depression in low-income women: Development of the Mamás y Bebés/Mothers and Babies Course. *Cognitive and Behavioral Practice*, 14, 70–83. http:// dx.doi.org/10.1016/j.cbpra.2006.04.021
- Murray, L., & Carothers, A. D. (1990). The validation of the Edinburgh Post-natal Depression Scale on a community sample. *The British Journal of Psychiatry*, 157, 288–290. http://dx.doi.org/10.1192/bjp.157.2 .288

- Mustillo, S., Krieger, N., Gunderson, E. P., Sidney, S., McCreath, H., & Kiefe, C. I. (2004). Self-reported experiences of racial discrimination and Black-White differences in preterm and low-birthweight deliveries: The CARDIA Study. *American Journal of Public Health*, 94, 2125– 2131. http://dx.doi.org/10.2105/ajph.94.12.2125
- O'Campo, P., Caughy, M. O., & Nettles, S. M. (2010). Partner abuse or violence, parenting and neighborhood influences on children's behavioral problems. *Social Science & Medicine*, 70, 1404–1415. http://dx .doi.org/10.1016/j.socscimed.2009.11.036
- O'Hara, M. W., & McCabe, J. E. (2013). Postpartum depression: Current status and future directions. *Annual Review of Clinical Psychology*, 9, 379–407. http://dx.doi.org/10.1146/annurev-clinpsy-050212-185612
- Pascoe, E. A., & Smart Richman, L. (2009). Perceived discrimination and health: A meta-analytic review. *Psychological Bulletin*, 135, 531–554. http://dx.doi.org/10.1037/a0016059
- Pearlstein, T., Howard, M., Salisbury, A., & Zlotnick, C. (2009). Postpartum depression. American Journal of Obstetrics and Gynecology, 200, 357–364. http://dx.doi.org/10.1016/j.ajog.2008.11.033
- Pew Research Center. (2017). How the U.S. Hispanic population is changing. Washington, DC: Author. Retrieved from https://www.pewresearch .org/fact-tank/2017/09/18/how-the-u-s-hispanic-population-is-changing/
- Pew Research Center. (2019). Race in American 2019. Washington, DC: Author. Retrieved from https://www.pewsocialtrends.org/2019/04/09/ race-in-america-2019/
- Pico-Alfonso, M. A., Garcia-Linares, M. I., Celda-Navarro, N., Blasco-Ros, C., Echeburúa, E., & Martinez, M. (2006). The impact of physical, psychological, and sexual intimate male partner violence on women's mental health: Depressive symptoms, posttraumatic stress disorder, state anxiety, and suicide. *Journal of Women's Health*, 15, 599–611. http:// dx.doi.org/10.1089/jwh.2006.15.599
- Rabin, R. F., Jennings, J. M., Campbell, J. C., & Bair-Merritt, M. H. (2009). Intimate partner violence screening tools: A systematic review. *American Journal of Preventive Medicine*, 36, 439–445.e4. http://dx.doi .org/10.1016/j.amepre.2009.01.024
- Ramey, S. L., Schafer, P., DeClerque, J. L., Lanzi, R. G., Hobel, C., Shalowitz, M., . . . the Community Child Health Network. (2015). The Preconception Stress and Resiliency Pathways Model: A multi-level framework on maternal, paternal, and child health disparities derived by community-based participatory research. *Maternal and Child Health Journal*, 19, 707–719. http://dx.doi.org/10.1007/s10995-014-1581-1
- Raudenbush, S. W., Bryk, A. S., & Congdon, R. (2017). HLM 7.03 for windows [Computer software]. Skokie, IL: Scientific Software International.
- Rich-Edwards, J. W., Kleinman, K., Abrams, A., Harlow, B. L., McLaughlin, T. J., Joffe, H., & Gillman, M. W. (2006). Sociodemographic predictors of antenatal and postpartum depressive symptoms among women in a medical group practice. *Journal of Epidemiology and Community Health*, 60, 221–227. http://dx.doi.org/10.1136/jech.2005 .039370
- Robertson, E., Grace, S., Wallington, T., & Stewart, D. E. (2004). Antenatal risk factors for postpartum depression: A synthesis of recent literature. *General Hospital Psychiatry*, 26, 289–295. http://dx.doi.org/ 10.1016/j.genhosppsych.2004.02.006
- Rodriguez, M. A., Heilemann, M. V., Fielder, E., Ang, A., Nevarez, F., & Mangione, C. M. (2008). Intimate partner violence, depression, and PTSD among pregnant Latina women. *Annals of Family Medicine*, 6, 44–52. http://dx.doi.org/10.1370/afm.743
- Rodríguez, M. A., Valentine, J., Ahmed, S. R., Eisenman, D. P., Sumner, L. A., Heilemann, M. V., & Liu, H. (2010). Intimate partner violence and maternal depression during the perinatal period: A longitudinal investigation of Latinas. *Violence Against Women*, *16*, 543–559. http:// dx.doi.org/10.1177/1077801210366959
- Rosenthal, L., & Lobel, M. (2018). Gendered racism and the sexual and reproductive health of Black and Latina Women. *Ethnicity & Health*.

Advance online publication. http://dx.doi.org/10.1080/13557858.2018 .1439896

- Schulz, A. J., Gravlee, C. C., Williams, D. R., Israel, B. A., Mentz, G., & Rowe, Z. (2006). Discrimination, symptoms of depression, and selfrated health among African American women in Detroit: Results from a longitudinal analysis. *American Journal of Public Health*, 96, 1265– 1270. http://dx.doi.org/10.2105/AJPH.2005.064543
- Shaffer, C. F. (2002). Factors influencing the access to prenatal care by Hispanic pregnant women. *Journal of the American Academy of Nurse Practitioners*, 14, 93–96. http://dx.doi.org/10.1111/j.1745-7599.2002 .tb00097.x
- Sherin, K. M., Sinacore, J. M., Li, X. Q., Zitter, R. E., & Shakil, A. (1998). HITS: A short domestic violence screening tool for use in a family practice setting. *Family Medicine*, 30, 508–512.
- Silverman, M. E., & Loudon, H. (2010). Antenatal reports of prepregnancy abuse is associated with symptoms of depression in the postpartum period. *Archives of Women's Mental Health*, 13, 411–415. http://dx.doi.org/10.1007/s00737-010-0161-7
- Surkan, P. J., Peterson, K. E., Hughes, M. D., & Gottlieb, B. R. (2006). The role of social networks and support in postpartum women's depression: A multiethnic urban sample. *Maternal and Child Health Journal*, 10, 375–383. http://dx.doi.org/10.1007/s10995-005-0056-9
- Tamis-LeMonda, C. S., & Kahana-Kalman, R. (2009). Mothers' views at the transition to a new baby: Variation across ethnic groups. *Parenting: Science and Practice*, 9(1–2), 36–55. http://dx.doi.org/10.1080/ 15295190802656745
- Tanner Stapleton, L. R., Dunkel Schetter, C., Dooley, L. N., Guardino, C. M., Huynh, J., Paek, C., . . . the Community Child Health Network. (2016). The Community Child Health Network Life Stress Interview: A brief chronic stress measure for community health research. *Anxiety*, *Stress & Coping*, 29, 352–366. http://dx.doi.org/10.1080/10615806 .2015.1058368
- Torres, L., Driscoll, M. W., & Voell, M. (2012). Discrimination, acculturation, acculturative stress, and Latino psychological distress: A moderated mediational model. *Cultural Diversity and Ethnic Minority Psychology*, 18, 17–25. http://dx.doi.org/10.1037/a0026710
- Torres, L., & Ong, A. D. (2010). A daily diary investigation of Latino ethnic identity, discrimination, and depression. *Cultural Diversity and Ethnic Minority Psychology*, 16, 561–568. http://dx.doi.org/10.1037/ a0020652
- Treyvaud, K., Lee, K. J., Doyle, L. W., & Anderson, P. J. (2014). Very preterm birth influences parental mental health and family outcomes seven years after birth. *The Journal of Pediatrics*, 164, 515–521. http:// dx.doi.org/10.1016/j.siny.2013.10.008
- Tucker, J., & Lowell, C. (2016). National snapshot: Poverty among women & families, 2015. Washington, DC: National Women's Law Center.
- U.S. Census Bureau. (2009). *Poverty thresholds*. Retrieved from https:// www.census.gov/data/tables/time-series/demo/income-poverty/ historical-poverty-thresholds.html
- Valentine, J. M., Rodriguez, M. A., Lapeyrouse, L. M., & Zhang, M. (2011). Recent intimate partner violence as a prenatal predictor of maternal depression in the first year postpartum among Latinas. *Archives* of Women's Mental Health, 14, 135–143. http://dx.doi.org/10.1007/ s00737-010-0191-1
- Vliegen, N., Casalin, S., & Luyten, P. (2014). The course of postpartum depression: A review of longitudinal studies. *Harvard Review of Psychiatry*, 22(1), 1–22. http://dx.doi.org/10.1097/hrp.000000000000013
- Williams, D. R., Neighbors, H. W., & Jackson, J. S. (2003). Racial/ethnic discrimination and health: Findings from community studies. *American Journal of Public Health*, 93, 200–208. http://dx.doi.org/10.2105/ajph .93.2.200
- Williams, D. R. (1997). Race and health: Basic questions, emerging directions. Annals of Epidemiology, 7, 322–333. http://dx.doi.org/10.1016/ S1047-2797(97)00051-3

556

- World Health Organization. (2015). Postnatal care for mothers and newborns: Highlights from the World Health Organization 2013 guidelines. Retrieved from https://www.who.int/maternal_child_ adolescent/publications/WHO-MCA-PNC-2014-Briefer_A4.pdf
- Yonkers, K. A., Ramin, S. M., Rush, A. J., Navarrete, C. A., Carmody, T., March, D., . . . Leveno, K. J. (2001). Onset and persistence of postpartum depression in an inner-city maternal health clinic system. *The American Journal of Psychiatry*, 158, 1856–1863. http://dx.doi.org/10 .1176/appi.ajp.158.11.1856
- Zayas, L. H., Jankowski, K. R., & McKee, M. D. (2003). Prenatal and postpartum depression among low-income Dominican and Puerto Rican women. *Hispanic Journal of Behavioral Sciences*, 25, 370–385. http:// dx.doi.org/10.1177/0739986303256914
- Zayas, L. H., & Sampson, M. (2014). Perinatal Depression Treatments for U.S. Latinas: A Review of Research Findings. In S. Lara-Cinisomo & K. L. Wisner (Eds.), *Perinatal depression among Spanish-speaking and Latin American women* (pp. 65–82). New York, NY: Springer. http:// dx.doi.org/10.1007/978-1-4614-8045-7_5